Author's response to reviews

Title: Somatic mutation analysis of MYH11 in breast and prostate cancer

Authors:

Pia Alhopuro (pia.alhopuro@helsinki.fi)
Auli Karhu (auli.karhu@helsinki.fi)
Robert Winqvist (robert.winqvist@oulu.fi)
Kati Waltering (kati.waltering@uta.fi)
Tapio Visakorpi (tapio.visakorpi@uta.fi)
Lauri A Aaltonen (lauri.aaltonen@helsinki.fi)

Version: 4 Date: 8 September 2008

Author's response to reviews:

To whom it may concern:

Thank you for notification of acceptance of our article for publication in BMC Cancer. We have reformatted the manuscript according to your requests. Abbreviations paragraph has been added to the manuscript. References have been formatted as requested by the Editorial Team (commas inserted between the surname and forename initial).

Sincerely,

Lauri A Aaltonen, MD, PhD, Professor